Author's response to reviews

Title: An investigation of the smoking behaviours of parents before, during and after the birth of their children in Taiwan

Authors:

Shu-Fang Shih (annssshih@nhri.org.tw)
Likwang Chen (likwang@nhri.org.tw)
Chi Pang Wen (cwengood@nhri.org.tw)
Wei-Chih Yang (weichih@nhri.org.tw)
Yaw-Tang Shih (stone@tche.org.tw)

Version: 6 Date: 22 January 2008

Author's response to reviews: see over
Dear Dr. Good,

We would like to thank you for giving us one more opportunity to further refine our research article entitled “An investigation of the smoking behaviours of parents before, during and after the birth of their children in Taiwan” (MS: 6380226851564452). We also appreciate the reviewers for helping us to further improve our article.

In this revision, we have incorporated all suggestions for minor essential revisions from the first reviewer. We have also checked our manuscript for typographical and grammatical by a professional editor and ourselves. We have checked the format of our paper, too.

We hope that you find this version is now acceptable for publication. Thank you for all your help.

Yours sincerely,

Likwang Chen, Ph.D.

Associate Investigator
Center for Health Policy Research and Development
National Health Research Institutes
Taiwan